Author's response to reviews

Title: Antibiotics for bronchiectasis exacerbations in children: rationale and study protocol for a randomised placebo-controlled trial

Authors:

Anne B Chang (annechang@ausdoctors.net)
Colin F Robertson (colin.robertson@rch.org.au)
Keith Grimwood (k.grimwood@uq.edu.au)
Peter P van Asperen (peter.vanasperen@health.nsw.gov.au)
Andrew C Wilson (Andrew.Wilson@health.wa.gov.au)
Kerry-Ann F O&#8217;Grady (k.ogrady@uq.edu.au)
Theo P Sloots (t.sloots@uq.edu.au)
Paul J Torzillo (pault@med.usyd.edu.au)
Emily J Bailey (e.bailey1@uq.edu.au)
Gabrielle B McCallum (Gabrielle.McCallum@menzies.edu.au)
Ian B Masters (brent_masters@health.qld.gov.au)
Catherine A Byrnes (c.byrnes@auckland.ac.nz)
Helen M Buntain (helen_buntain@health.qld.gov.au)
Ian M Mackay (ian.mackay@uq.edu.au)
Peter S Morris (Peter.Morris@menzies.edu.au)
Mark D Chatfield (mark.chatfield@menzies.edu.au)

Version: 7 Date: 16 August 2012

Author's response to reviews: see over
16th Aug 2012

Dear Editors

Re: Submission of manuscript for protocol “Randomised placebo-controlled trial on antibiotics for bronchiectasis exacerbations in children: rationale and protocol”

Thank you for the review and accepting our manuscript for publication in Trials. We have re-formatted the article and changed the title as required.

Yours sincerely

Anne Chang on behalf of co-authors